Author’s response to reviews

Title: Health-related quality of life as an endpoint in oncology phase I trials: a systematic review

Authors:

Frédéric Fiteni (fredericfiteni@gmail.com)
Isabelle Le Ray (isabelle.le.ray@normalesup.org)
Ahmad Ousmen (aousmen@chu-besancon.fr)
Nicolas Isambert (NIsambert@cgfl.fr)
Amélie Anota (aanota@chu-besancon.fr)
Franck Bonnetain (franck.bonnetain@univ-fcomte.fr)

Version: 3 Date: 31 Mar 2019

Author’s response to reviews:

1. Please represent authors' names using their initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Full names have been replaced by initials

2. We note that you have not included an acknowledgements section. If you have no acknowledgements please put ‘Not Applicable’ in this section.

Acknowledgement section has been added

3. Please move the Declarations to come between the Abbreviations and References.

Declarations have been put between abbreviations and references
4. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

The statement has been added

5. Please remove the PRISMA checklist, as it is no longer needed at this stage.

PRISMA checklist has been removed

6. Please include a Figure Legends heading above the Figure Legends.

A heading has been added